Author's response to reviews

Title: Late Presentation of Sorafenib-Associated Rash: A Case Report.

Authors:

Thomas Sarkodie (tsarkodie@gmail.com)
Paul Ross (paul.ross@gstt.nhs.uk)

Version: 3 Date: 18 April 2010

Author's response to reviews:

1. Title changed to Late presentation of sorafenib-associated rash: A case report
2. Abstract restructured into: Introduction, Case presentation, Conclusion
3. BCLC staging of HCC moved from middle of abstract to end of abstract
4. Bibliography of patient elaborated
5. Individual authors’ contribution summarised
6. Table 4 omitted
7. Timing of rash presented as a graph
8. CT scans provided